Author's response to reviews

Title: Improving adherence to ante-retroviral treatment for people with harmful alcohol use in Kariobangi, Kenya through participatory research and action

Authors:

Caleb J Othieno Dr (cjothieno@uonbi.ac.ke)
Anne A Obondo Dr (nnobondo2@gmail.com)
Mathai Muthoni Dr (Mathai@web.de)

Version: 4 Date: 2 July 2012

Author's response to reviews:

June 12 2012

The Editor,

Dear Sir,

MS: 1507665207639703 Improving adherence to ante-retroviral treatment for people with harmful alcohol use in Kariobangi, Kenya through participatory research and action

Thank you for the reviewers’ comments. I have made the corrections in the background section as suggested by referee 1.

1. Paragraph 1 Line 4: I have deleted the words “of sero-prevalence” from the sentence and it now reads “... higher prevalence rates ranging from...”

2. Paragraph 3 Line 2: I have deleted the word “the” from the sentence and it now reads “Although controversial ... “

Sincerely,

Caleb Othieno